Author's response to reviews

Title: Fatigue during treatment for hepatitis C virus: results of self-reported fatigue severity in two Phase IIb studies of simeprevir treatment in patients with hepatitis C virus genotype 1 infection

Authors:

Jane Scott Dr (jscott51@its.jnj.com)
Kathleen Rosa Dr (quantpsyc@aol.com)
Min Fu Dr (mfu1@its.jnj.com)
Karin Cerri Dr (kcerri@its.jnj.com)
Monika Peeters Dr (mpeeter7@its.jnj.com)
Maria Beumont-Mauviel Dr (mbmauvie@its.jnj.com)
Stefan Zeuzem Dr (zeuzem@em.uni-frankfurt.de)
Donna M Evon Dr (donna_evon@med.unc.edu)
Leen Gilles Dr (lgilles@its.jnj.com)

Version: 5
Date: 23 July 2014

Author's response to reviews: see over
Fatigue during treatment for hepatitis C virus: results of self-reported fatigue severity in two Phase IIb studies of simeprevir treatment in patients with hepatitis C virus genotype 1 infection

<table>
<thead>
<tr>
<th>Editorial request</th>
<th>Response</th>
</tr>
</thead>
</table>
| Ethical approval - Please update the Methods section of your manuscript to include the name of the ethics committee that approved your study. We recommend the following format: ?This study was approved by the ethics committee of [xxx] Hospital [or University].? If your study was conducted at more than five centres, you may wish to include a list of all of the ethics committee names as a supplementary file. | • The methods section of the manuscript has been revised to include this information.  
• In addition, the acknowledgments have been revised to acknowledge the clinical investigators for the TMC435 C205 and C206 studies.  
• A list of the investigators and Ethics Committees who approved the protocols is provided in the supplemental information for this article. |